Author's response to reviews

Title: UK Waiting Time Targets In Lung Cancer Treatment: Are They Achievable? Results Of A Prospective Tracking Study

Authors:

Mohan P Devbhandari (mohandev@hotmail.com)
Sing Y Soon (SingYang.Soong@smuht.nwest.nhs.uk)
Pauline Quennell (Pauline.Quennell@smuht.nwest.nhs.uk)
Phil Barber (Phil.Barber@smuht.nwest.nhs.uk)
Piotr Krysiak (Piotr.Krysiak@smuht.nwest.nhs.uk)
Rajesh Shah (Rajesh.Shah@smuht.nwest.nhs.uk)
Mark T Jones (Mark.Jones@smuht.nwest.nhs.uk)

Version: 3 Date: 6 December 2006

Author's response to reviews:

Mohan Devbhandari
Flat 28 Fern House
Amberly Drive
Wythenshawe, Manchester
United Kingdom
Tel: +44 161 9803100 Fax: +44 1612912685
E mail: mohandev@hotmail.com

6 December 2006

Vipin Zamvar
Editor in Chief Journal of Cardiothoracic Surgery
Royal Infirmary of Edinburgh
Edinburgh

Dear Mr Zamvar,

Subject: Original article: UK Waiting Time Targets In Lung Cancer Treatment: Are They Achievable? Results Of A Prospective Tracking Study

Thank you very much for accepting the above article for publication in the journal. We are sorry for the delay in reformatting the paper. We have made the following changes as suggested

Formatting changes requested and made

Title - Please use sentence case in the title, both in the manuscript file and the submission system (i.e. remove all unnecessary capitals). done

Affiliations - Each affiliation should be written in full and be provided in the following format: department, institute, city, country. Please do not use full capitals for country names, and please do not include full-stops in abbreviated country names. done
Can we ask you to link each author to their corresponding affiliation with the use of superscript numbers. done

Qualifications - Please remove the authors' qualifications (e.g. PhD / MD) from the manuscript and submissions system. done
Thanking you very much for your help. Kind regards.

Yours truly,

Mohan Devbhandari
Corresponding author